Author's response to reviews

**Title:** Managing a locally advanced malignant thymoma complicated by nephrotic syndrome - a challenging scenario.

**Authors:**

Daren CY Teoh (dcy_teoh@hotmail.com)
Ahmed El-Modir (elmodir@hotmail.com)

**Version:** 4  **Date:** 23 January 2008

**Author's response to reviews:** see over
23/1/08

Prof Kidd,
Editor,
JMCR.

Re: 2nd resubmission after Peer-review - point-by-point alterations made in response

PC comments: Make clear that acute renal impairment may be contributed by other factors.

We have done this in the case report section to make this clear.

Many thanks.

Yours truly,

Daren Teoh/ Ahmed El-Modir.